**Author’s response to reviews**

**Title:** Inclusion of joint laxity, recurrent patellar dislocation, and short distal ulnae as a feature of Van Den Ende-Gupta syndrome: A case report

**Authors:**

Mohammad Al-Qattan (moqattan@hotmail.com)

Doaa Andejani (dandejani@hotmail.com)

Nadia Sakati (nasakati@gmail.com)

Khushnooda Ramzan (kramzan@kfshrc.edu.sa)

Faiq a Imtiaz (fahmad@kfshrc.edu.sa)

**Version:** 3 **Date:** 19 Jan 2018

**Author’s response to reviews:**

MGTC-D-17-00267R2

Inclusion of joint laxity, recurrent patellar dislocation, and short distal ulnae as a feature of Van Den Ende-Gupta syndrome: A case report

BMC Medical Genetics

Dear Editor,

Thank you for considering the manuscript. We have revised it as follows:

1. The individual contributions of each author of the manuscript should be specified in the Authors’ Contributions section, separately, by their initials.

   **Response:** Done. See lines 208-216

2. There is still some writing in some of the X-rays, please remove this, as it appears to be a name (e.g. "aa Mesfer Hendi"). Words such as "upright" and "standing" can be left in the images, as they are not identifying.
Response: Fig 2b was corrected

3. The CARE checklist can be removed at this stage. Please include a sentence in the Case presentation stating that the CARE guidelines were followed in this report.

Response: Done. See lines 90-91